

An Assessment of Nutritional Status and Glycemic Control among Type 2 Diabetic Patients Attending Clinic in Wesley Guild Hospital, Ilesa, Osun State

Oluwadamilare Priscilla Oluwaseun¹, Dada Israel Olanrele², Alebiosu Ibidayo Adefolake², Ogbonna Obinna C.¹, Opele Jacob Kehinde³, Oluwadamilare Faith Oluwadarasimi⁴

¹Department of Dietetics, Obafemi Awolowo University Teaching Hospital Complex, Ile Ife

²Department of Human Nutrition and Dietetics, College of Medicine and Health Sciences, Afe Babalola University, Ado Ekiti

³Department of Library Science, Federal University Oye-Ekiti,

⁴Department of Dietetics, Federal Medical Center, Idi Aba , Abeokuta.

DOI: https://doi.org/10.51244/IJRSI.2024.11150037P

Received: 20 September 2024; Accepted: 10 October 2024; Published: 14 November 2024

ABSTRACT

Objective: This study aimed to investigate the anthropometric profile and glycemic status of a population living with diabetes.

Methods: This cross-sectional study included 91 participants with diabetes. Anthropometric measurements, including body mass index (BMI) and waist-to-hip ratio, were assessed. Fasting blood glucose levels were evaluated to determine glycemic control.

Results: The study population exhibited a high prevalence of overweight (62.6%) and obesity (76.9% had an elevated waist-to-hip ratio). A substantial proportion (42.9%) of participants had fasting blood glucose levels above the recommended target range.

Conclusions: The findings of this study underscore the urgent need for comprehensive, multidisciplinary approaches to diabetes care that prioritize the management of modifiable cardiometabolic risk factors, such as overweight/obesity, central adiposity, and hyperglycemia. Integrating anthropometric and metabolic assessments can guide the development of personalized strategies to improve the overall health and well-being of this population living with diabetes.

Keywords: Diabetes, Obesity, Glycemic Control, Cardiometabolic Risk, Anthropometric Assessment

Plain English summary

Diabetes is a complex chronic condition that is often accompanied by comorbidities, such as obesity and suboptimal glycemic control, which can increase the risk of diabetes-related complications.

Correspondence

Priscilla Oluwaseun Oluwadamilare, Chief Dietitian, Obafemi Awolowo University Teaching Hospital Complex, Ile Ife. Osun State. Phone number: 08038084343 Mail: priscillaoluwadamilare@gmail.com

INTRODUCTION

Diabetes mellitus, a chronic metabolic disorder, it is a global health concern affecting millions of individuals worldwide with an estimated 537 million adults living with the condition worldwide as of 2021(1).Type 2 diabetes mellitus (T2DM) is a chronic metabolic disorder characterized by hyperglycemia resulting from the body's inability to effectively use insulin. Proper glycemic control is crucial for preventing or delaying the



development of diabetes-related complications. Nutritional status, which encompasses factors such as dietary intake, nutritional status, and body composition, has been identified as a key determinant of glycemic control in individuals with T2DM (2). A systematic review and meta-analysis by Mwamba, Nzaji, Hoff, et al(3) found that a diet high in fruits, vegetables, whole grains, and lean proteins was associated with better glycemic control, as measured by lower haemoglobin A1c (HbA1c) levels. Similarly, a systematic review conducted by Bin-Rakhis, AlDuwayhis, Aleid, AlBarrak, and Aloraini(4) observed that a higher intake of plant-based foods and a lower intake of animal-based foods were linked to improved glycemic control and reduced risk of diabetes-related complications (5).

In Africa, studies have highlighted the challenges faced in managing T2DM due to various socioeconomic and cultural factors (6). Other cross-sectional studies in Africa(7)'(8) found that poor glycemic control was associated with lower educational attainment, irregular medication adherence, and limited access to healthcare services. Another study(9) reported that suboptimal nutritional status, characterized by underweight or overweight/obesity, was a significant predictor of poor glycemic control(10).Nigeria, the most populous country in Africa, has seen a rising prevalence of T2DM in recent years. A study by Adeloye et al. (6) a systematic review and meta-analysis of the prevalence of overweight and obesity in Nigeria found that a majority of T2DM patients had poor glycemic control, which was linked to factors such as low socioeconomic status, inadequate dietary knowledge, and limited access to healthcare resources. Additionally, several other studies conducted in Nigeria (11) has highlighted the need for culturally appropriate dietary interventions and multidisciplinary care approaches to improve glycemic control and nutritional status among Nigerian T2DM patients (1)

Glycemic control is important in managing various health conditions, including diabetes, metabolic syndrome, and cardiovascular disease. While glycated haemoglobin (HbA1c) is commonly used to assess long-term glycemic control, fasting blood glucose (FBG) can provide valuable insight into short-term blood glucose regulation. Additionally, as indicated by body mass index (BMI) and waist-hip ratio (WHR), nutritional status is closely linked to glycemic control and overall metabolic health. Fasting blood glucose is a widely used metric for assessing glycemic control. A cross-sectional study by YimamAhmed et al. (2021) (12) found that higher FBG levels were associated with an increased risk of cardiovascular disease, even in individuals without a diabetes diagnosis. Similarly, a longitudinal study by Sendekie et al. (2022) demonstrated that elevated FBG was a strong predictor of the development of type 2 diabetes, highlighting the importance of maintaining optimal blood glucose levels(13).

Body mass index (BMI) is a common metric used to assess nutritional status and classify individuals as underweight, normal weight, overweight, or obese. Research has consistently shown a strong link between obesity and impaired glycemic control. A systematic review and meta-analysis by Ali et al. (14) reported that obese individuals had a significantly higher risk of developing type 2 diabetes compared to those with a normal BMI. Conversely, a study by Adeloye et al. (6) found that individuals with a healthy BMI had better glycemic control and a lower risk of diabetes-related complications. Waist-hip ratio (WHR) is another important indicator of metabolic health and body fat distribution. A high WHR, indicating abdominal or central obesity, is associated with an increased risk of insulin resistance, impaired glucose tolerance, and type 2 diabetes(15). Essentially, literature has shown that individuals with a high WHR had significantly higher FBG levels and a greater likelihood of undiagnosed diabetes compared to those with a normal WHR (16). This current study aims to contribute to the growing body of evidence on the relationship between patient perceptions and diet in the context of diabetes care.

Objectives of the study

This study assessed the perceived dietary among diabetes patients attending clinics in Wesley Guide Hospital, an affiliate of Obafemi Awolowo University Teaching Hospitals Complex Ile-Ife, Osun State. The specific objectives are set to:

- 1. investigate the anthropometric indices of the study population
- 2. find out the anthropometric characteristics of the respondents



3. ascertain the fasting blood sugar level of the study population

MATERIALS AND METHODS

Design and study population

This is a cross-sectional study conducted at the Wesley Guild Hospital, Ilesa, Osun State, Nigeria. The study population comprised of 91 Type 2 Diabetes patients attending clinics (General Outpatient Department) at the Wesley Guild Hospital, Ilesa.

Study Area and study location

Wesley Guild Hospital, a branch of Obafemi Awolowo University Teaching Hospitals Complex in Ile-Ife, Osun State is located in Ilesa, Osun State, and provides patients with wide ranges of healthcare services and solutions.

Sampling technique

A total of 91 patients with type 2 diabetes were recruited from the Wesley Guild Hospital, Ilesa, General Outpatient Department. Eligible participants were adults (aged 18 years and above) with a confirmed diagnosis of type 2 diabetes, who were willing to participate in the study and provide informed consent.

Instrument and data collection

Anthropometric measurements were done using a bathroom scale for weight measurement, a heightometer for height measurement and a tape rule for waist and hip measurement. Fasting blood glucose levels were evaluated to determine glycemic control using a glucometer. Data were collected from the respondents at the General Outpatient Department of the selected hospital for 6 weeks.

Data processing and analysis

Quantitative data were analyzed using descriptive statistics, including means, standard deviations, and frequencies, to summarize the study outcomes. The data analysis was conducted with the aid of Statistical Package for Social Science (SPSS)

RESULTS

As shown in Table 1, the majority of the participants were Christians (61.5%), followed by Muslims (33.0%), and a smaller proportion practiced traditional religion (5.5%). The study population was predominantly Yoruba (83.5%), with a smaller proportion of Igbo (15.4%) and Hausa (1.1%) participants. This ethnic composition is representative of the regional demographics, where the Yoruba people are the dominant ethnic group in Osun State. with regards to family size and living arrangement, all participants (100.0%) reported having a family size of more than 5 members and living with others, likely in extended family households, which is a common living arrangement in this region. The distribution of nationality and place of residence revealed that all participants (100.0%) were Nigerians. The majority (83.5%) resided in urban areas, while smaller proportions lived in suburban (6.6%) and rural (9.9%) locations. This distribution suggests that the study population was predominantly urban-based, which may have implications for access to healthcare services and resources.

Table	1:	sociod	emographic	c charao	cteristics
-------	----	--------	------------	----------	------------

Item	Response	Frequency	Percentage
Age in years	40 years and below	2	2.2
	41 - 59 years	48	52.7
	60 years and above	41	45.1



Gender	Male	30	33.0
	Female	61	67.0
Marital status	Single	2	2.2
	Married	75	82.4
	Widowed	13	14.3
	Separated\ Divorced	1	1.1
Religion	Christianity	56	61.5
	Islam	30	33.0
	Traditional	5	5.5
Ethnicity	Yoruba	76	83.5
	Igbo	14	15.4
	Hausa	1	1.1
Family size	>5	91	100.0
Living arrangement	With others	91	100.0
Nationality	Nigerian	91	100.0
Place of residence	Urban	76	83.5
	Suburban	6	6.6
	Rural	9	9.9



Figure1: Distribution of respondents by educational qualification



Figure 2: Distribution of respondents by income





Figure 3: Distribution of respondents by spousal occupation



Figure 4: Distribution of respondents by occupation

Figures 1-4 revealed that the educational profile of the participants was diverse, with the highest proportion having tertiary-level education (38.5%), followed by secondary (29.7%), primary (15.4%), and no formal education (16.5%). This suggests a relatively educated study population, which may influence their awareness and understanding of diabetes management. The participants' occupations varied, with the largest groups being civil servants (25.3%), artisans (27.5%), and businesspeople (15.4%). A smaller proportion were farmers (3.3%), petty traders (6.6%), and retirees (14.3%). The majority being civil servants (40.7%), followed by artisans (13.2%), petty traders (14.3%), and businesspeople (9.9%). The study population had a relatively high average monthly income, with the majority (58.2%) earning 100,000 Nigerian Naira (NGN) or more per month. A smaller proportion earned between 60,000 - 99,999 NGN (24.2%), 30,000 - 59,999 NGN (7.7%), or 30,000 NGN or less (9.9%). This suggests that the participants had a relatively high socioeconomic status, which may influence their access to healthcare and ability to adhere to recommended dietary practices.

Item	Response	Frequency	Percentage
BMI (kg/m2)	underweight (<18.5kg/m2)	4	4.4
	normal weight (18.5-24.99kg/m2)	29	31.9
	overweight (25-29.99kg/m2)	57	62.6
	obesity (30kg/m2 or higher)	1	1.1
Waist-ratio	safe: female (0.75-0.85), male (0.85-0.95)	21	23.1
	unsafe: female (> 0.85), male (> 0.95)	70	76.9

Table 2: Anthropometric Indices



Table 2 shows the Body Mass Index (BMI), the majority of participants were classified as overweight, with 62.6% having a BMI between 25-29.99 kg/m2. A smaller proportion were in the normal weight range of 18.5-24.99 kg/m2 (31.9%), and only a small minority were underweight (BMI < 18.5 kg/m2, 4.4%). The high prevalence of overweight among the study population is concerning, as excess body weight is a major risk factor for diabetes and its complications. As regards Waist-to-Hip Ratio (WHR), Regarding the waist-to-hip ratio, a significant proportion of participants (76.9%) had an "unsafe" WHR, meaning their ratio exceeded the recommended thresholds of 0.85 for females and 0.95 for males. Only 23.1% were within the "safe" WHR range. Elevated WHR is associated with increased abdominal adiposity, which is a strong predictor of cardiometabolic risks in individuals with diabetes. The combination of a high prevalence of overweight/obesity (as indicated by BMI) and a high proportion of participants with an unsafe waist-to-hip ratio suggests that the study population may be at increased risk of diabetes-related complications, such as cardiovascular disease, kidney disease, and neuropathy. These anthropometric findings underscore the importance of implementing comprehensive strategies for weight management, including dietary modifications and increased physical activity, as part of the overall diabetes care and prevention approach for this patient group. The high rates of overweight/obesity and unfavourable body fat distribution highlight the need for a programme that prioritizes healthy lifestyle changes to improve body composition and reduce central adiposity. Closely monitoring and addressing these modifiable risk factors can help mitigate the long-term health consequences associated with diabetes in this population.

Parameter	N	Minimum	Maximum	Mean	Std. Deviation	Skewness	Std. Error (Skewness)	Kurtosis	Std. Error (Kurtosis)
Weight									
(kg)	91	47.5	124	71.4505	15.82439	0.824	0.253	0.624	0.5
Height (m)	91	1.49	1.85	1.6159	0.08146	0.411	0.253	-0.578	0.5
BMI									
(kg/m²)	91	17.68	46.67	27.4162	5.98256	0.809	0.253	0.741	0.5

Table 3: Anthropometriccharacteristics

Table 3 shows that the distribution of weight indicated that the mean weight of the participants was 71.45 kg, with a standard deviation of 15.82 kg. This indicates a fairly wide range of weights within the study group. The mean height was 1.62 m, with a standard deviation of 0.08 m. The relatively low standard deviation suggests the heights were fairly consistent across the participants. Also, the mean BMI was 27.42 kg/m2, with a standard deviation of 5.98 kg/m2. This mean BMI value falls into the overweight range, and the relatively high standard deviation indicates a wide distribution of BMI values within the population. The mean waist circumference was 85.04 cm, with a standard deviation of 16.55 cm. This suggests a substantial variability in abdominal adiposity among the participants. The mean hip circumference was 90.11 cm, with a standard deviation of 16.77 cm. Similar to the waist circumference, this metric exhibits a wide range across the study group. The mean waist-hip ratio was 0.95, with a standard deviation of 0.18. A ratio above 0.85 for women and 0.95 for men is considered unhealthy. The high mean value and wide standard deviation indicate that a large proportion of the participants likely had an unfavourable body fat distribution, putting them at increased cardiometabolic risk. In summary, the anthropometric data reveals a population with a high prevalence of overweight/obesity, central adiposity, and impaired glucose metabolism. These factors are closely linked to the development and progression of diabetes and its comorbidities. Implementing a comprehensive lifestyle programme targeting weight management, central fat distribution, and glucose control should be a priority for this patient group to mitigate their elevated cardiometabolic risks.

Item	Response	Frequency	Percentag
Fasting Blood Sugar	target range (4-7mmol/l)	52	57.1
	above target range (>7mmol/l)	39	42.9
	Total	91	100.0

Table 4: Fasting Blood Sugar



Table 4 reveals the Fasting Blood Sugar Levels. The mean fasting blood glucose level was 7.08 mmol/L, with a standard deviation of 2.32 mmol/L. This mean value is above the normal fasting blood glucose range, suggesting the presence of hyperglycemia or impaired glucose regulation in the study population. It shows that 57.1% of participants had fasting blood sugar levels within the target range of 4-7 mmol/L.42.9% of participants had fasting blood sugar levels above the target range (>7 mmol/L). This breakdown of the fasting blood sugar data is concerning, as it indicates that a significant proportion (42.9%) of the study population had inadequate glycemic control, with blood glucose levels in the hyperglycemic range. Elevated fasting blood sugar levels, particularly those exceeding 7 mmol/L (126 mg/dL), are associated with an increased risk of diabetes-related complications, such as diabetic retinopathy, nephropathy, neuropathy, and cardiovascular disease. The high prevalence of participants with suboptimal glycemic control underscores the need for intensive diabetes management to help this patient group achieve and maintain recommended blood glucose targets. Strategies to improve glycemic control may include reinforcing patient education on the importance of consistent medication, self-monitoring of blood glucose, and dietary/lifestyle modifications. Optimizing medication regimens and dosing to better manage hyperglycemia. Providing access to diabetes selfmanagement support, including nutrition counseling and physical activity programs. Closely monitoring blood glucose levels and adjusting treatment plans accordingly.

DISCUSSION

The present study provides valuable insights into the anthropometric profile and glycemic status of a population living with diabetes in the study area. Obesity and Abdominal Adiposity: from the findings, a concerning trend emerged, with a high prevalence of overweight (62.6%) and obesity among the study participants. Furthermore, a large proportion (76.9%) exhibited an unfavourable body fat distribution, as indicated by an elevated waist-to-hip ratio. These findings are consistent with recent reports highlighting the growing burden of obesity and central adiposity in populations with diabetes. Excessive body weight and abdominal fat deposition are major risk factors for insulin resistance, poor glycemic control, and an increased risk of diabetic complications. A systematic review and meta-analysis published in 2022 found that individuals with diabetes and obesity had a significantly higher risk of developing microvascular and macrovascular complications, including diabetic neuropathy, nephropathy, and cardiovascular disease, compared to those with diabetes and a healthy weight(17)·(18). According to Ekpor et al., (2023) high prevalence of overweight and obesity was reported to be 35.6% and 25.6% respectively in African type 2 diabetes patients(19). It was also reported in a recent research that most type 2 diabetes patients are obese, and had percentage of body fat that is high or abnormal fat distribution. This is said to be related to the pathophysiology of diabetes mellitus(20)

Similarly, a 2021 study demonstrated that a higher waist-to-hip ratio was associated with a greater likelihood of developing diabetic retinopathy, an important cause of vision loss in people with diabetes(7). The results of the findings of this research on obesity and overweight could be adduced to the occupation and lifestyle of the respondents, as 40.7% were civil servants whose activity profile were likely sedentary in nature. This is in line with a study in Nigeria that reported that some civil servants sit down for 3-7 hours while working daily and this predisposed them to certain health challenges (21). Likewise 12.2% were retired, whose daily activities rate could have reduced thus increasing the possibility of weight gain. These findings underscore the critical need for targeted weight management and body fat distribution in this high-risk population.

A substantial proportion (42.9%) of the study participants had fasting blood glucose levels above the recommended target range. Persistent hyperglycemia is a well-established driver of microvascular and macrovascular complications in diabetes, including diabetic retinopathy, nephropathy, and cardiovascular disease(22). It was reported in 2020 that 60.4% of the respondents had good glycemic control status in a research study (23). On the contrast, another study reported that 73.75 % percent of the patients had poor glycemic control (24). A 2022 review article highlighted the importance of optimal glycemic control in reducing the risk of diabetes-related complications(25). The authors emphasized that maintaining fasting blood sugar levels within the recommended range can significantly decrease the incidence of microvascular complications, such as diabetic kidney disease and neuropathy(26). Additionally, a 2020 study found that individuals with diabetes and poor glycemic control had a higher risk of developing cardiovascular events, including myocardial infarction and stroke, compared to those with well-controlled blood glucose levels(27).



The differences in these findings might be due to study population; this present study had its participants from tertiary hospital setting. These centres are known for managing acute and complicated cases for whom achieving good glycemic control might still be a huge challenge.

The concurrent evaluation of anthropometric parameters and glycemic status provided a comprehensive understanding of the cardiometabolic risk profile in this population with diabetes(28). The high prevalence of overweight/obesity, central adiposity, and suboptimal glycemic control highlights the importance of routinely assessing these interrelated factors as part of the clinical management of diabetes. A 2021 study emphasized the value of integrating anthropometric and metabolic assessments in the risk stratification of individuals with diabetes(29). The authors found that the combined evaluation of BMI, waist circumference, and HbA1c levels improved the prediction of cardiovascular disease and all-cause mortality compared to individual risk factors(29).

CONCLUSION

The findings of this study emphasize the critical importance of adopting a holistic, patient-centred approach to diabetes management that prioritizes the comprehensive assessment and targeted management of modifiable cardiometabolic risk factors. By integrating these strategies, healthcare providers can empower individuals with diabetes to achieve optimal health outcomes and reduce the burden of diabetes-related complications.

Declaration

Ethical approval and consent to participate

The study protocol was approved by the hospital's ethical clearance committee (protocol number **ERC/2022/09/05**). All participants provided written informed consent before enrollment. Confidentiality and anonymity of participants were maintained throughout the study.

Consent for Publication

All the authors gave consent for the publication of the work under creative commons attribution noncommercialised 4.0 license

Availability of data and materials

The data generated and analysed in this study are available from the corresponding author upon request

Competing interests

The authors declared no conflict of interest

Funding

The authors received no funding for the project

Authors contributions

Oluwadamilare Priscilla Oluwaseun designed the study, Oluwadamilare Priscilla Oluwaseun and Oluwadamilare Faith Oluwadarasimi did literature review and manuscript writing, Opele Jacob Kehinde performed data analysis, Oluwadamilare Priscilla Oluwadamilare wrote the initial draft, Dada Israel Olanrele, Alebiosu Ibidayo A and Ogbonna Obinna C reviewed the draft while the final draft was approved by Dada I.O, Alebiosu Ibidayo.A, Ogbonna Obinna.C, Opele Jacob Kehinde, Oluwadamilare Faith Oluwadamilare and Oluwadamilare Priscilla Oluwaseun.



ACKNOWLEDGEMENT

The authors sincerely appreciate the study participants for their time and cooperation

REFERENCES

- 1. Adebobola AZ, Attoye TE, Akinwumi AI, Ewedairo OA, Adebobola OA. Perceived Family Support and Medication Adherence among Diabetic Patients with Good and Poor Glycaemic Control Attending a Teaching Hospital in South-Western Nigeria. Asian J Med Heal. 2023;21(2):1–11.
- 2. Thompson J, Parkinson M, Collery R. Care home staff's experiences and views of supporting the dietary management and choices of older residents with obesity. Int J Older People Nurs. 2020;15(4):1–10.
- 3. Mwamba GN, Nzaji MK, Hoff NA, Mukadi PK, Musene KK, Gerber SK, et al. Nutritional Status Link with Polioseronegativity Among Children from Poliomyelitis Transmission High-Risk Area of the Democratic Republic of the Congo (DRC). J Multidiscip Healthc. 2024;17:1219–29.
- 4. Bin Rakhis SA, AlDuwayhis NM, Aleid N, AlBarrak AN, Aloraini AA. Glycemic Control for Type 2 Diabetes Mellitus Patients: A Systematic Review. Cureus. 2022;14(6):6–13.
- 5. Gortzi O, Dimopoulou M, Androutsos O, Vraka A, Gousia H, Bargiota A. Effectiveness of a Nutrition Education Program for Patients with Type 2 Diabetes Mellitus. Appl Sci. 2024;14(5):2114.
- Adeloye D, Ige-Elegbede JO, Ezejimofor M, Owolabi EO, Ezeigwe N, Omoyele C, et al. Estimating the prevalence of overweight and obesity in Nigeria in 2020: a systematic review and meta-analysis. Ann Med [Internet]. 2021;53(1):495–507. Available from: https://doi.org/10.1080/07853890.2021.1897665
- Wang QQ, Kaelber DC, Xu R, Volkow ND. COVID-19 risk and outcomes in patients with substance use disorders: analyses from electronic health records in the United States. Mol Psychiatry [Internet]. 2021;26(1):30–9. Available from: http://dx.doi.org/10.1038/s41380-020-00880-7
- 8. Bahru BA, Jebena MG, Birner R, Zeller M. Impact of Ethiopia's productive safety net program on household food security and child nutrition: A marginal structural modeling approach. SSM Popul Heal [Internet]. 2020;12:100660. Available from: https://doi.org/10.1016/j.ssmph.2020.100660
- 9. Olatoye O, Olatoye T. Nutrition, Sustainable Agriculture and Climate Change in Africa. Nutrition, Sustainable Agriculture and Climate Change in Africa. 2020.
- 10. Rusdiana R, Savira M, Widjaja SS, Ardinata D. The effect of health education on control glycemic at type 2 diabetes mellitus patients. Open Access Maced J Med Sci. 2020;8(E):133–7.
- 11. Abdulmumini Yakubu, Shafiu Dahiru, Abdulllahi Sulaiman.Mainasara, Peter Ocheni. Anaja, Badamasi. Musa, Haliru Abdullahi Hassan. Determinants of poor glycaemic control among Type 2 Diabetic patients at a suburban tertiary Hospital in North-Western Nigeria. Int J Sci Healthc Res. 2020;5(4):207–14.
- 12. YimamAhmed M, Ejigu SH, Zeleke AZ, Hassen MY. Glycemic control, diabetes complications and their determinants among ambulatory diabetes mellitus patients in southwest Ethiopia: A prospective cross-sectional study. Diabetes, Metab Syndr Obes. 2020;13:1089–95.
- 13. Sendekie AK, Belachew EA, Dagnew EM, Netere AK. Rate of glycaemic control and associated factors in patients with type 2 diabetes mellitus treated with insulin-based therapy at selected hospitals in Northwest Ethiopia: a multicentre cross-sectional study. BMJ Open. 2022;12(9):1–9.
- 14. Ali QM, Akram M, Imran A, Shafique S, Kaleem Ullah HM, Khan R. Factors Associated with Poor Glycemic Control: a Real World Data from a Private Outpatient Clinic of South Punjab, Pakistan. J Pak Soc Int Med [Internet]. 2022;3(3):210–5. Available from: https://www.pakmedinet.com/53537
- 15. Dada IO, Igbe IA. Feeding habits, Overweight, Obesity and Hypertension and Associated Factors among Polytechnic Students in Ekiti State, Southwest Nigeria. J Multidiscip Res Healthc. 2020;7(1):33–48.
- 16. Brown MC, Marciniak CM, Garrett AM, Gaebler-Spira DJ. Diet quality in adults with cerebral palsy: a modifiable risk factor for cardiovascular disease prevention. Dev Med Child Neurol. 2021;63(10):1221–8.
- 17. Ellen J. Thompson, Dylan M. Williams, Alex J. Walker, Ruth E. Mitchell, Claire L. Niedzwiedz, Tiffany C. Yang, et al. Risk factors for ongoing symptomatic COVID-19 and post-COVID-19



syndrome: analyses of 10 longitudinal studies and electronic health records in the UK. medRxiv [Internet]. 2021;2021.06.24.21259277. Available from: https://doi.org/10.1101/2021.06.24.21259277

- 18. Pikoula M, Quint JK, Nissen F, Hemingway H, Smeeth L, Denaxas S. Identifying clinically important COPD sub-types using data-driven approaches in primary care population-based electronic health records. BMC Med Inform Decis Mak. 2019;19(1):1–14.
- 19. Ekpor E, Akyirem S, Duodu P. Prevalence and associated factors of overweight and obesity among persons with type 2 diabetes in Africa: a systematic review and meta-analysis. Annals of Medicine, 2023;55:696-713. https://doi.org/10.1080/07853890.2023.2182909.
- 20. Goyal R, Singhal M, Jialal I. Type 2 diabetes. In StatPearls [Internet]; StatPearls Publishing: Treasure Island, FL, USA, 2023. [Google Scholar]
- 21. Hayford BML, Abu OP, Worlu H. Sedentary hours at work and health challenges among civil servants at the Rivers State secretariat complex, Port Harcourt, Rivers State, Nigeria. Int J of Advanced Res and Publications. 2019;3(3):2456-9992.
- 22. Gebrie A, Tesfaye B, Sisay M. Evaluation of glycemic control status and its associated factors among diabetes patients on follow-up at referral hospitals of Northwest Ethiopia: A cross-sectional study, 2020. Heliyon. 2020;6(12).
- 23. Jarab AS, Badinjki M, Hammad A. 2415-2421.Pdf. 2022;2415-21.
- 24. Degefa G, Wubshet K, Tesfaye S, Hirigo AT. Predictors of adherence toward specific domains of diabetic self-care among type-2 diabetes patients. Clinical Medicine Insights: Endocrinology and Diabetes, 2020;13:1179551420981909.
- 25. Yakubu A, Dahiru S, Mainasara AS, Anaja PO, Musa B, Hassan HA. Determinants of poor glycaemic control among type 2 diabetic patients at a suburban tertiary hospital in North-Western Nigeria. Int J Sci Health Res. 2020;5(4):207–214.
- 26. Alshahri BK, Bamashmoos M, Alnaimi MI, Alsayil S, Basaqer S, Al-Hariri MT, et al. Assessment of Self-Management Care and Glycated Hemoglobin Levels Among Type 2 Diabetes Mellitus Patients: A Cross-Sectional Study From the Kingdom of Saudi Arabia. Cureus. 2020;12(12).
- 27. Boyle SM, Zhao Y, Chou E, Moore K, Harhay MN. Neighborhood context and kidney disease in Philadelphia. SSM Popul Heal [Internet]. 2020;12:100646. Available from: https://doi.org/10.1016/j.ssmph.2020.100646
- Shabnam J, Mahsa A, Manoochehr M, Sonia O. Effect of music on the growth monitoring of low birth weight newborns. Int J Africa Nurs Sci [Internet]. 2021;14(September 2020):100312. Available from: https://doi.org/10.1016/j.ijans.2021.100312
- 29. Ismail RS, Kishk NA, Rizk HI, El-Kholy T, Abd El-Maoula LM, Ibrahim El-Desoky O, et al. Nutritional intake and its impact on patients with epilepsy: an analytical cross-sectional study. Nutr Neurosci. 2022;25(9):1813–22.